Author’s response to reviews

Title: Sentinel lymph node biopsy is unsuitable for routine practice in younger female patients with unilateral low-risk papillary thyroid carcinoma

Authors:

Ou Huang (ou_huang@medmail.com.cn)
Wei Li Wu (weili5566@sina.com)
Ou Chen Wang (woc099@gmail.com)
Jie You (wzclouds@163.com)
Quan Li (liquansky@gmail.com)
Du ping Huang (oncology2004@163.com)
Xiao Qu Hu (drhxj@163.com)
Jin Miao Qu (oncologist_qjm@sina.com)
Cun Jin (doctor_kingson@sina.com)
You Qun Xiang (tom02@sina.com)
Kai Yang (yangkai1006@yahooomcom.cn)
Shu Mei Zhou (surzsm@163.com)
Xue Min Chen (mqj16@126.com)
Yi Fei Pan (13506641535@126.com)
Guo Long Guo (gu quilong@sina.com)
Xiao Huag Zhang (zhangxiaohua991180@126.com)

Version: 7 Date: 16 August 2011

Author’s response to reviews: see over
Dear editor:

This is the final edition of my article. I have added the support funds in the part of “competing interests” in red.

We have agreed to pay the article processing charge by the funds. However, the term of funds only supported the article which is accepted after September 20. Could you do me a favor to delay the accepted time to that time or later?

Thank you very much. Please give me a reply to my request.

Good luck!